Author's response to reviews

Title: Coronary Artery Aneurysm: Case Report

Authors:

  Jeffrey E Everett (jeverett@utmck.edu)
  Harold M Burkhart (burkhart.harold@mayo.edu)

Version: 2 Date: 26 November 2007

Author's response to reviews: see over
November 26, 2007

We appreciate the comments from the editors and have made revisions to address their concerns. The following is a point-by-point listing of the revisions:

1. Line 2 of case presentation the word from correctly replaces form
2. Discretionary revisions made to address other forms of treatment including covered stents and medical management. Also added was a discussion of management of proximal left main aneurysm, and how to gain proximal control. The clinical presentation was modified to include the presence of murmurs when there is an associated fistula. Three references were added to support the additional comments.